Author's response to reviews

Title: A comparison of indices of glucose metabolism in five black populations: Data from Modeling the Epidemiologic Transition Study (METS).

Authors:

Yacoba Atiase (yacobat@gmail.com)
Kathryn Farni (kfarni@luc.edu)
Jacob Plange-Rhule (jprhule@gmail.com)
Amy Luke (aluke@luc.edu)
Pascal Bovet (bovet.pascal@gmail.com)
Terrence G Forrester (terrence.forrester@uwimona.edu.jm)
Estelle V Lambert (vicki.lambert@uct.ac.za)
Naomi S Levitt (naomi.levitt@uct.ac.za)
Stephanie Kliethermes (skliethermes@luc.edu)
Guichan Cao (gcao@luc.edu)
Ramon A Durazo-Arvizu (rdurazo@luc.edu)
Richard S Cooper (rcooper@luc.edu)
Lara R Dugas (ldugas@luc.edu)

Version: 5 Date: 31 July 2015

Author's response to reviews:

August 2nd, 2015

Diana Marshall
Senior Managing Editor
BMC Public Health

Dear Dr. Marshall

Thank you for allowing us to resubmit our manuscript “4486119841601026- A comparison of indices of glucose metabolism in five black populations: Data from Modeling the Epidemiologic Transition Study (METS).” We have made the edit to the abstract in a revised submission.

Thank you for your consideration.

Yours truly,

Lara Dugas, PhD, MPH